Thursday, 31 August 2006

To the Editor;

Re: Manuscript 9143046671051557

Thank you for the opportunity to publish in BMC Medical Research Methodology. As per your email of the 29th of August the following changes have been made to the manuscript as requested.

1) Country has been added to the affiliation details.

2) Manuscript layout has been changed to the following order; Competing interests, Authors' contributions, Acknowledgements.

3) Authors contributions has been changed to Authors’ contributions.

Sincerely

EA Nelson